Author’s response to reviews

Title: Cancer suspicion in general practice, urgent referral and time to diagnosis: A population-based cross-sectional study

Authors:

Henry Jensen (henry.jensen@alm.au.dk)
Marie Louise Tørring (mlt@alm.au.dk)
Frede Olesen (fo@alm.au.dk)
Jens Overgaard (jens@oncology.dk)
Peter Vedsted (p.vedsted@alm.au.dk)

Version: 3 Date: 30 May 2014

Author’s response to reviews:

Dear Editor,

BMC Cancer

As requested we have included the figure legend at the end of the document. Also, we have embedded the four tables in the manuscript as well.

Kind Regards

Henry Jensen

PhD research Fellow, MHSc

Corresponding Author